Author's response to reviews

Title: Otitis media in children vaccinated with either 7-valent or 10-valent pneumococcal conjugate vaccines: a cross sectional comparison

Authors:

  Amanda J Leach (Amanda.Leach@menzies.edu.au)
  Christine Wigger (Christine.Wigger@menzies.edu.au)
  Ross Andrews (Ross.Andrews@menzies.edu.au)
  Mark Chatfield (mark.chatfield@menzies.edu.au)
  Heidi Smith-Vaughan (heidi.smith-vaughan@menzies.edu.au)
  Peter S Morris (peter.morris@menzies.edu.au)

Version: 7  Date: 22 July 2014

Author's response to reviews:

Table 2 has had the Title changed from '451 PCV7' to '444 PCV7', and '444 PHiD-CV10' to '451 PHiD-CV10'.

Table 3b has had the Risk Difference for "Any sibling with history of "runny ears"" changed from -5% [-12 to 3] to 5% [-3 to 12].